Author’s response to reviews

Title: Gender-based differences in letters of recommendation written for ophthalmology residency applicants

Authors:

Fei Lin (flin@mednet.ucla.edu)
Soo Oh (osk1403@ucla.edu)
Lynn Gordon (gordon@jsei.ucla.edu)
Stacy Pineles (pineles@jsei.ucla.edu)
Jamie Rosenberg (jarosenb@montefiore.org)
Irena Tsui (itsui@jsei.ucla.edu;irena.tsui@va.gov)

Version: 2 Date: 12 Dec 2019

Author’s response to reviews:

Dear Editor Fagelson:

Thank you for the positive feedback and opportunity to correct our manuscript entitled, "Gender-based differences in letters of recommendation written for ophthalmology residency applicants". Author contributions, funding, and abbreviations have been updated. Please see attached a clean copy for review.

Best, Irena Tsui